Widespread and Persistent Deposition of Iron Formations for Two Billion Years

Jena E Johnson¹ and Peter H Molnar²

¹Department of Earth and Environmental Sciences, University of Michigan, Ann Arbor MI 48109

²Department of Geological Sciences, University of Colorado, Boulder CO 80309

Corresponding author: Jena E Johnson (jenaje@umich.edu)

Key Points:

- A subset of iron formations, known as Algoma-type, can be interpreted as accreted deeper marine sediment atop oceanic crustal rocks
- We present a preservation-corrected iron formation temporal record
- The scaled iron formation record suggests widespread, potentially ocean-wide, and continued deposition of iron formations from 3.8 to 1.7 Ga

This is the author manuscript accepted for publication and has undergone full peer review but has not been through the copyediting, typesetting, pagination and proofreading process, which may lead to differences between this version and the Version of Record. Please cite this article as doi: 10.1029/2019GL081970

This article is protected by copyright. All rights reserved.

Abstract

Composed of chemical precipitates rich in iron and silica, Precambrian iron formations from marine sedimentary records may reveal biogeochemical processes over the first half of Earth history. The limited record of early Archean rock suggests that preservation biases the iron formation record. Like ophiolites, which provide a sparse record of past ocean floor, iron formations deposited on oceanic crust ought also be rare and preserved only when accreted onto cratons. To correct for potential preservation bias, we scaled masses of iron formations to the areal extent of basement rock of similar age and found that the resultant record is consistent with persistent deposition of iron formations across much of the deep ocean for two billion years. Widespread and long-term iron formations imply that ferrous iron was available in the deep ocean for billions of years and that the requisite (bio)geochemical mechanisms to produce iron formations were present by 3.8 Ga.

Keywords

Archean ocean chemistry; temporal record of iron formations; early Earth iron cycle

Plain Language Summary

Why did the ancient ocean deposit an increasing volume of iron over time, until these 'Iron Formations' largely disappeared? Does this pattern reflect a growing titration of oxygen, a surge in iron from mantle superplumes, or the prerequisite of developing continental crust? Or is this distribution simply from the diminishing record of ancient rocks? We normalize the iron formation record over time by scaling it to crustal preservation, and the normalized record suggests that there may have been widespread and unchanging deposition of these iron formations across the ancient oceans for two billion years – implying that the (bio)chemical process responsible for making iron formations operated from ~3.8 to ~1.8 billion years ago.

1. Introduction

With no modern analogues for the early ocean, the study of ancient marine iron- and silica-rich chemical precipitates offers insights into how this alien sea operated (e.g., Klein, 2005; Konhauser et al., 2017). These peculiar precipitates, containing ~30% iron and ~50% silica, were deposited primarily between ~3.8 and ~1.7 billion years ago (Ga) in sediment-starved marine settings (Trendall, 2002; Klein, 2005). Known as 'iron formations', here abbreviated IFs, they are studied both as records of the Archean and Proterozoic ocean chemistry and as sources of economically important iron ores (Clout and Simonson, 2005). Although the mechanism that formed the original IFs remains enigmatic, the transport and concentration of reduced iron in the ocean, away from continental weathering and hydrothermal sources, indicate that environmental oxygen levels were much lower than today (e.g., Cloud, 1968; Bekker et al., 2010).

The uneven IF record, with large volumes near the Archean-Proterozoic boundary at 2.5 Ga, has prompted many hypotheses suggesting various controls on IF deposition. Cloud (1973) proposed that early microbial oxygen producers originally were in balance with reduced iron, suppressing the release of oxygen, but then these microbes overcame physiological barriers to completely titrate the iron and allow a rise in oxygen. Recent dating, however, shows that the apparent ~2.5-Ga IF peak precedes the ca. 2.4-2.3 Ga major rise in oxygen ("Great Oxidation Event" or GOE) (Luo et al., 2016; Gumsley et al., 2017). More recently, others have considered the IF temporal distribution as regulated by either limited stable depositional repositories (James, 1983; Trendall, 2002; Simonson, 2003) or insufficient ferrous iron, with influxes of ferrous iron —perhaps from upwelling hydrothermal fluids or mantle plume events— stimulating major IF deposition (Holland, 1973; Morris, 1993; Barley et al., 1997; Isley and Abbott, 1999; Konhauser et al., 2017). Others have suggested the confluence of both depositional basins and fluxes of iron together modulated the IF record (Clout and Simonson, 2005; Bekker et al., 2010, 2014).

An obstacle to evaluating hypotheses for the temporal distribution of IFs is the long-recognized underestimation of Archean iron formations (Gole and Klein, 1981; James, 1983). Both subduction and deformation leading to "dismemberment", combined with erosion and the dwindling exposure of the older rock record on modern Earth, have greatly diminished the original extent of the earliest iron formations (Gole and Klein, 1981). Additionally, the growth of continental crust and continental shelves over time not only affected the volume of iron formation that might have been deposited, but also provided environments that can preserve IFs (Trendall, 2002; Simonson, 2003). Although previous publications and correct for them, apart from a recent illustrative analysis using just the record from North America, New Zealand and the Caribbean (Peters and Husson, 2018).

Another complicating factor is the exclusion of small, igneous rock-associated IFs from compilations of IFs over geologic time (Bekker et al., 2010, 2014). Because lithological hosts of IFs vary, differing categorizations of IFs have been proposed (James, 1954; Gross, 1965; Kimberley, 1978; Beukes, 1980). In the most widely used of these, Gross (1965, 1980) defined Algoma-type IFs as associated with volcanic rock in deep-water settings, below wave base, if not at depths of kilometers, and Superior-type IFs as affiliated with strata deposited on continental shelves and slopes. Most Algoma-type IFs consist of relatively thin (centimeter to tens of meter) layers associated with greenstone belts of igneous and sedimentary rock assemblages, whereas many of the thicker (often hundreds of meters thick) and more extensive Superior-type IFs accumulated on cratonic margins (Bekker et al., 2014; Konhauser et al., 2017). Despite the Algoma-type IFs having similar geochemistry to the Superior-type IFs (Gole and Klein, 1981; Trendall, 2002; Beukes and Gutzmer, 2008), Bekker et al. (2010, 2014) excluded Algoma-type IFs in recent IF compilations, with the argument that these record local environments associated with volcanic or hydrothermal inputs rather than representing large-scale ocean chemistry.

We contend that Algoma-type IFs must be included in IF compilations to understand the significance of iron formations as recorders of early Earth marine biogeochemistry. Outcrops of Algoma-type iron formations, often deposited on basalt in ocean basins, share features associated with ophiolites in more modern tectonic settings. Although we do not equate them with ophiolites, we suggest that many Algoma-type IFs to be remnants of ocean floor sediment, later accreted onto continents along with the underlying basalt rock, as Rosing et al. (1996) interpreted for the earliest known IF (Isua). When smoothly functioning, subduction removes most oceanic crust, so that many obducted ophiolites can be seen as accidents (Moores, 1982; Dilek and Furnes, 2014). The ocean crust and uppermost mantle comprising many ophiolites formed in back arc settings, in some cases with island arcs built on them, and were later emplaced on continental crust as that crust was subducted beneath the arc (Moores, 1982; Dilek and Furnes, 2014). Regardless of how ophiolites have been emplaced, however, they offer an incomplete record of past seafloor. Whether or not basaltic crust capped rigid plates of lithosphere, as in classic plate tectonics, convective overturn in the Archean mantle likely carried oceanic crust back into the mantle. Accordingly, preserved oceanic crust with Algoma-type iron formations should similarly be viewed as an incomplete record of past ocean floor, including volcanic arcs in some cases. From this perspective, their abundance in the geologic record today places only a qualitative constraint on the extent to which iron and silica were deposited in Archean time. Despite these limitations, Algoma-type IFs may actually provide our best record of the sediment, and the geochemical record that it contains, deposited on the Archean ocean floor.

The Superior-type IFs, associated with voluminous deposition on the margins of cratons, have a clear relationship to the preservation of continental crust (e.g., Trendall, 2002). Cratonic margins, including convergent settings where flexure of the lithosphere may have dropped strata below wave base (e.g., Hoffman, 1987, 1988; Fedo and Eriksson, 1996), provide preservational advantages for these IFs, even when buried or folded in collisional orogenies. Continental crust is predicted to escape subduction (e.g., McKenzie, 1969; Molnar and Gray, 1979), which

efficiently banishes thinner, denser oceanic crust back into the mantle. Moreover, thick layers of buoyant Archean upper mantle caused its overlying continental crust to be less susceptible to both subduction and deformation than crust that caps younger lithosphere (Jordan, 1988). If Earth wanted to preserve intact sedimentary layers, it could choose no better place than Archean cratons, including their margins. The evidence for widespread development of continental margins near the Archean-Proterozoic boundary (Bickle et al., 1975; Beukes, 1987; Bradley, 2008; Campbell and Davies, 2017) may account for the exceptional examples of Superior-type IFs from ~2.7 to 2.4 Ga.

2. Removing Preservational Biases from the IF Record

To determine whether IF deposition underwent an increase at ~2.5 Ga or the major IF volumes around this time reflect the growth of appreciable and better preserved continental shelf area (Figure 1), the IFs need to be scaled to a measure of the preservational biases against the older rock record. Although Archean continental crust might have been nearly as extensive as crust today (e.g., Armstrong, 1981), most of that crust has been altered or buried, and can provide no record of early iron formations. To estimate how biased the present-day record of IFs is, we need to weight estimates of the initial IF masses (estimated using the observed thicknesses and areal extents of each IF, in gigatons (Gt), Table S1) against the amount of preserved rock from that time period. Using K/Ar ages, but adjusted for Ar loss using Rb/Sr ages and initial values of ⁸⁷Sr/⁸⁶Sr, Hurley and Rand (1969) measured areas of exposed continental basement rock through time from the Americas, Africa, Europe, India, Australia, and Antarctica; they could not access adequate information for the former Soviet Union or China. Because of the dependence on K/Ar ages, some of their ages were likely incorrect, but the considerable amount of data used and the binning into 450-Myr windows compensates for such errors. Hurley and Rand's compilation therefore contains accurate dates for the regions and time bins that they considered. The areas of exposed continental basement are subject to the same biases that the ancient IFs have: limitations from subduction, tectonic reworking, and exposure. Therefore, using Hurley and Rand's

measurements removes much of the preservational biases embedded in the raw temporal record of IF volume. We assigned each IF to the appropriate 450-Myr bin assigned by Hurley and Rand (Figure 1B) and divided the mass estimate by the total measured area of exposed continental crust from that age bin. For IFs older than 3.15 Ga, we assumed the preservation of crust increased linearly from nil at 4.0 Ga to 1,065,600 km² at 3.15 Ga, the value that Hurley and Rand gave for the interval from 3.15 to 2.7 Ga (Figure 1B). We assign no physical meaning to the dimensions, Gt/km², of these ratios, and merely use them to compare accumulation amounts over a period of time during which preservation has drastically changed.

We applied this preservation scaling to previous compilations of IF tonnage estimates over time from 4 Ga to 1 Ga (Isley and Abbott, 1999; Huston and Logan, 2004; Bekker et al., 2014). The masses for each IF are order-of-magnitude estimates originally from James (1983), who acknowledged that the fragmentary nature of the geologic record suggests "that the time distribution pattern may be more related to accidents of preservation than to a time-controlled cycle of deposition." Successive compilations, by Isley (1995) and Isley and Abbott (1999), Huston and Logan (2004), and then Bekker et al. (2010, 2014), added IFs and refined mass estimates. For virtually every major IF, masses are estimated to only one significant figure, and therefore uncertainties are at least a factor of 3 of those estimated masses. Continually improving geochronology allowed us to refine some IF age constraints (Figure 1A, Table S1). Scaling these mass estimates to preserved continental area alters the impression of an IF peak at ~2.7-2.4 Ga, as Figures 2 and S1 show with histograms of the raw masses of IFs over time compared to preservation-scaled IF amounts.

The approach taken here necessarily ignores some major iron formations. In particular, those in China and the former Soviet Union were excluded from Figure 2 as we cannot correct them for preservational bias using Hurley and Rand's areal compilation. From China, we excluded the small 1700 ± 100 Ma Chuanlinggou IF (Wan et al., 2003; Bekker et al., 2014), the ~100 Gt

Liaohe Formation constrained to 1990 \pm 60 Ma Liaohe Formation (Luo et al., 2004; Bekker et al., 2014), and the more important 2700 \pm 100 Ma Anshan Liaoning deposit with 10,000 Gt of IF mass (Table S1) (Bekker et al., 2014; Hao et al., 2017). Canfield et al. (2018) recently revised the mass of the ca. 1400 Ma Xiamaling Formation to 520 Gt. The imprecisely dated (between 2700 Ma and 2080 Ma) Krivoy Rog IF in the former Soviet Union is estimated to have contained 50,000-100,000 Gt of IF (Kulik and Korzhnev, 1997; Isley and Abbott, 1999; Bekker et al., 2014). The Belozyorsky-Konsky IF in Ukraine, with an estimated age of 3267 \pm 26 Ma (Isley, 1995), may have originally contained 1000 Gt of IF (Isley, 1995; Huston and Logan, 2004). The Kursk Magnetic Anomaly (KMA) suggests a large unexposed IF is in the Ukrainian shield, with estimated iron reserves at least 13 times larger than Krivoy Rog (Alexandrov, 1973). Similarly, Launay et al. (2018) inferred from a large magnetic anomaly that the total volume of the ca. 2.1 Ga West African Ijil IF is 350,000 km³, and therefore ~1,000,000 Gt, more than 3 times larger than any other IF (Figure 1). Notably, this IF accumulated after the GOE. Since we correct for biases of preservation and exposure by normalizing to exposed volume to scale for preservation.

The most recent estimates of Superior-type IF masses (Bekker et al., 2014) combined with those from Algoma-type IFs (Isley and Abbott, 1999; Huston and Logan, 2004) and other estimates (Table S1) yield a compilation of IF mass/area ratios from 4.0 to 1.0 Ga (Figure 3), evaluating the temporal record of reported IFs from the Eoarchean to the Mesoproterozoic Era. This preservation-scaled accumulation record is contextualized by additionally plotting oceanwide deposition of IF. Measured thicknesses of Algoma-type iron formations can be as thin as centimeters, but many are tens of meters (see SI). Therefore, an ocean floor covered by 25 meters of IF is a reasonable estimate to use as a representative thickness of ocean-wide, Algoma-type IFs in Figure 3. We follow Labrosse and Jaupart (2007) and assume that the mean age of the oceanic crust has changed little since Archean time, and therefore equaled the present-day mean age of ~80 Myr. Figure 3 includes the well-dated Chinese and former Soviet Union IFs,

corrected using the appropriate time bin from Hurley and Rand but shown with dashed lines to note that the preservation correction does not incorporate the areal record from these regions. The poorly dated Krivoy Rog IF, unexposed KMA, and large buried portion of the Ijil IF were again excluded from this figure.

3. Sustained Ocean-Wide IF Deposition

The bias-corrected consensus record (Figure 3B) is more consistent with continual iron- and silica-rich precipitate deposition over the entire ocean from ~3.8 Ga to ~1.7 Ga than with an abrupt increase around 2.5 Ga. The apparent ~2.5 Ga "peak" in IF deposition has previously been interpreted as an environmental signal. Correcting for preservation and exposure biases, however, indicates that this late Archean "peak" in IF may be an artifact due to the growth and increased preservation of continents and continental shelf areas (Figure 2). The normalized IF record in Figure 3B allow an equally plausible hypothesis that extensive long-term IF deposition occurred, with an imperfect geologic record capturing the deeper ocean record in instances of accreted oceanic crust (Algoma-type IFs) and at the margins of growing continental shelves (Superior-type IFs). This preservation-scaled temporal distribution of IF masses likewise suggests that influences of changing chemistry, e.g., hydrothermal/mantle plumes of iron (Isley and Abbott, 1999) or biology, e.g., related to the evolution of oxygenic photosynthesis and associated release of O₂ (Cloud, 1973), may have been overemphasized, if relevant at all, in IF deposition. The accumulation of iron-rich and silica-rich precipitates appears to have been an important oceanic process as far back as 3.5 Ga and possibly 3.8 Ga (Figure 3). Rather than viewing the IF record as reflecting oceanic (bio)chemical changes or the availability of stable depositional basins, we suggest this record could be viewed as a tectonically dismembered and fragmented record of continual deeper-water deposition of iron and silica precipitates.

If IF precipitation and deposition did occur across all deep ocean basins starting at 3.8 Ga, then the fluxes of iron and silica into seawater must equal or exceed the depositional fluxes of these

elements associated with IF formation. It is generally accepted that marine silica levels were higher in the Precambrian ocean due to the lack of silica-secreting organisms (Maliva et al., 2005). Iron enters the ocean today through continental weathering and hydrothermal systems, such as along mid-ocean ridge vents. Today these vents have variable iron concentrations, from 9 nanomolar to ~25 millimolar (Seyfried et al., 1991; Douville et al., 2002), and the total hydrothermal iron input has been estimated at $\sim 10^{10}$ kg (or 0.01 Gt) per year (Isley, 1995). In examining the mass balance of modern hydrothermal fluxes of iron, Isley (1995) determined that IF deposition would consume a large fraction of hydrothermal plume iron and that IF deposition near 2.5 Ga needed additional iron sources. Yet ancient hydrothermal vent fluids had a different balance of iron and sulfur, largely due to the lack of oceanic sulfate to reduce to hydrothermal sulfide (Walker and Brimblecombe, 1985; Kump and Seyfried, 2005). With circulating seawater containing low sulfate and little to no oxygen, hydrothermal fluids likely had Fe(II) concentrations as high as 80 mmol/kg (Kump and Seyfried, 2005). Continental weathering sources of iron could also have been higher than today without oxygen to 'trap' iron terrestrially in paleosols (e.g., Rye and Holland, 1998; Hao et al., 2017), but it is challenging to estimate Archean continental weathering sources of iron due to the uncertainty surrounding the development and emergence of continents (e.g., Bradley, 2008; Campbell and Davies, 2017).

To determine hydrothermal inputs of ferrous iron into the Archean ocean, we use estimates of the flux of seawater undergoing high-temperature hydrothermal circulation at modern spreading centers combined with estimated concentrations of iron in high-temperature Archean vent fluids. The latent heat released by the formation of crust at mid-ocean ridge systems today is transported by circulating seawater, with approximately 80% of the heat transported by water circulating at low temperatures along ridge flanks and 20% of the heat transported by high-temperature water circulating close to the ridge axis (German and Seyfried, 2014). This high-temperature fluid flux is estimated to be $0.17-2.93 \times 10^{13}$ kg/year using thallium isotopes (Nielsen et al., 2006). With higher Archean mid-ocean ridge temperatures (e.g., Sleep and Windley, 1982; Herzberg et al.,

2010), the high-temperature fluid flux was likely at least on the upper end of today's range, at $\sim 3 \times 10^{13}$ kg/year. With pre-GOE hydrothermal iron concentrations of 80 mmol/kg (Kump and Seyfried, 2005), hydrothermal fluids would introduce 0.134 Gt of Fe(II) per year (see SI).

Could this hydrothermal source flux supply sufficient iron for ocean-wide precipitation of IF precursors? Commonly used depositional rates for IFs come from the two of the largest Superiortype IF sequences: the Hamersley Group of Australia and the Transvaal Supergroup in South Africa. Assuming that 0.2 to 1.6 mm-scale laminae in Hamersley Group IFs represented varvelike annual depositional cycles, Trendall and Blockley (1970) estimated a high depositional rate for IFs of 227 m/Myr. Ewers and Morris (1981) later suggested an even higher rate of 893 m/Myr. Geochronological constraints of the Hamersley Group are consistent with a depositional rate as low as 7 m/Myr (Trendall et al., 2004); Trendall et al. preferred 180 m/Myr, not accounting for any depositional hiatuses. Others have proposed that many sedimentary hiatuses are recorded by the bedded cherts in IFs (Krapež et al., 2003). Much lower IF depositional rates in the Transvaal Supergroup, at 2-6 m/Myr, have been inferred from geochronological constraints (Barton et al., 1994; Pickard, 2003). If iron formations accumulated in deep water across Archean ocean basins as well as along continental margins, depositional rates of IF chemical precipitates may indeed have been quite low, similar to modern pelagic deposition accumulating ~1 to 6 m/Myr (Müller and Suess, 1979; Sugisaki, 1984). If average depositional rates of 1 m to 180 m/Myr characterized the entire ocean with a surface area of 4×10^{14} m² (78% of Earth's total surface area), endmember bounds on the IF iron sink would be ~0.36 to 65 Gt Fe per year.

The high depositional rates proposed by Trendall and others for continental shelf regions (Trendall and Blockley, 1970; Ewers and Morris, 1981) would not be consistent with extensive and long-term deposition, as Isley (1995) determined. Ocean-wide depositional rates similar to modern pelagic deposition, however, come closer to an equilibrium with the hydrothermal input

of Fe(II) calculated above. The lack of detrital material in IFs is also consistent with slow sedimentation in abyssal plains. It is possible that limited areas such as the Hamersley Basin and other continental shelves had high depositional rates, perhaps stimulated by Fe(II) upwelling, but those rapid rates could not be responsible for Algoma-type IFs spanning the Archean ocean. We can estimate a depositional rate for an average IF thickness, consisting of 30% Fe, covering the entire ocean basin in the period of time given by the mean age of the ocean floor. Ocean floor spanning 4×10^{14} m² and accumulating 25 m of IF sediment with a density of 3×10^3 kg/m³ (Trendall and Blockley, 1970) in 80 Myr would remove Fe from the ocean at 0.11 Gt Fe/yr (see SI), comparable to the hydrothermal Fe input of 0.134 Gt/yr calculated above. With additional iron fluxes from continental weathering and higher hydrothermal fluxes due to shallower and hotter Archean spreading centers (Sleep and Windley, 1982), the potential ocean-wide IF accumulation rate could reach 1 m/Myr (0.36 Gt Fe/yr) or even higher. Although depositional rates of 180-893 m/Myr are not feasible at this ocean-wide scale, a low IF sedimentation rate of ~1 m/Myr would also help explain the perplexing lack of organic matter in IFs (Klein and Beukes, 1989) by allowing extensive organic degradation to occur in slow-accumulating sediments during diagenesis. This post-depositional processing could also re-release ferrous iron back into seawater and act as an additional iron flux to balance the Archean iron cycle (Konhauser et al., 2005).

Sustained and ocean-wide accumulation of IF precipitates would make IFs almost certainly the largest sink for iron, and perhaps also for silica. Therefore (bio)geochemical processes precipitating these iron-and-silica precipitates would have acted as major controls on elemental cycling for over a billion years, from ~3.8 Ga to the ~2.3 Ga GOE, if not later. This apparent availability of iron throughout much of the Archean supports hypotheses of a long-term Archean ferruginous ocean (e.g., Poulton and Canfield, 2011). Persistent IF deposition, moreover, suggests that any organisms relying on iron for either enzymatic requirements or iron-based (iron-oxidizing or iron-reducing) metabolisms could have flourished over this entire ~1.5-Ga

time period. With no obvious change in IF deposition throughout this pre-GOE time, the IF temporal record appears less like a growing titration of oxygen and more consistent with over a billion years of mineralization processes unrelated to oxygen. Likewise, pervasive and widespread IF deposition implies that either IFs formed independently of biological influences or the necessary IF-generating organisms were prevalent beginning at ~3.8 Ga until the GOE.

After the GOE, deposition of iron formations continued for ~600 million years (Figure 3) but with marked differences in IF sedimentology. The thinly laminated, mud-sized chemical precipitates of pre-GOE IFs evolved to include thick units of cross-bedded, sand-sized granular clasts and ooids of iron-rich minerals (Simonson, 2003; Clout and Simonson, 2005). These sedimentological observations indicate that the depositional environment(s) for IFs moved from generally deeper sedimentation below wave base prior to the GOE to more shallow-water high-energy environments after the GOE, suggesting the locus of iron precipitation migrated to shallower water depths (Simonson, 2003; Clout and Simonson, 2005). The decline of sustained IF deposition after 2 billion years, beginning ~1.7 Ga, may relate to an increase in oceanic sulfate (Kump and Seyfried, 2005), sulfide (Canfield, 1998; Anbar and Knoll, 2002), or environmental oxygen (Huston and Logan, 2004; Holland, 2006; Sleep and Bird, 2008). Although few IFs are known between 1.7 and 1.0 Ga (Figure 3), IF deposition did not completely stop and strata from this period may yet yield more evidence of infrequent but significant iron precipitation, such as the ~1.4 Ga Xiamaling IF (Canfield et al., 2018).

4. Conclusions

The two-billion-year, preservation-corrected history of iron- and silica-rich precipitation and deposition (Figure 3) implies that IFs record a persistent and stable (bio)geochemical marine iron and silica cycle. The temporal pattern of IFs may not have been limited by depositional environments, nor by influxes of ferrous iron (Isley and Abbott, 1999; Trendall, 2002), and need not reflect an interplay of tectonic, chemical, and biological factors (Clout and Simonson, 2005;

Bekker et al., 2010). Instead, IFs may chronicle a widespread signal that was subsequently obscured and dismembered to form an incomplete record on Earth's surface today. Rather than focus on the ~2.5 Ga "peak" in IF volumes, which may be an artifact of continental growth and preservational biases, future research might do well to concentrate on understanding the features of pre-GOE IFs and how they changed post-GOE, including the reason(s) for (1) the deposition of IFs for ~1.5 billion years prior to the GOE, (2) the continued deposition of IFs after the GOE for hundreds of millions of years, and (3) the decrease, and eventual demise, of IFs after ~1.7 Ga.

Acknowledgements

We thank Norman Sleep and an anonymous reviewer for helpful reviews. Data used in figures can be found in Table S1 and in the U-Michigan Deep Blue online repository. We are grateful to Lang Farmer and Alexis Templeton for helpful discussions and to Jefferson Yarce for assistance with Matlab programming. We thank the University of Michigan and the Crafoord Foundation for funding support.



Figure 1: A. Mass estimates for IFs reported in the literature, with age uncertainty plotted as error bars. IFs that were reported but not given a mass or described as small/unknown in previous compilations were assigned a small mass of 0.01 Gt arbitrarily to indicate their presence. Note the abundance of low-mass but prevalent IFs throughout the Archean and early Proterozoic. **B.** Aggregated areal extent of continental basement rock (gray), after Hurley and Rand (1969) who binned areas into 450-million year periods, overplotted with the IF estimated masses (red) binned in 100 Myr intervals and using the maximum estimates for each formation solely to compare the general trend of the IF record against the temporal increase in preserved and exposed continental crust.



Figure 2. Histograms of IF estimated masses (Gt per 50 Myr bin) through time (above, gray), compared to histograms of masses scaled to preserved crustal area (below, black, Gt/km² per 50 Myr bin): **A.** From Isley (1995) and Isley and Abbott (1999); only to 1.5 Ga because that was the interval considered by Isley and Abbott. **B.** From Huston and Logan (2004). **C.** From Bekker et al. (2014). Note how preservation-scaling increases the apparent importance of earlier IFs.



Figure 3. Iron formation (IF) accumulation over time, in 50-Myr bins, using previous mass estimates of Isley (1995), Huston and Logan (2004), Bekker et al. (2014), and others (Table S1) and scaled by crustal preservation (see text). **A.** Assembled record shown with a linear scale as in Figure 2. The prominent peak at ~2.8 Ga is an artifact of its slightly higher mass estimate and should be disregarded because mass estimates are uncertain by a factor of 3. **B.** Log-transformed IF record, suggestive of persistent accumulation of IFs from ~3.8 Ga to ~1.7 Ga. The mass/area ratio expected for 25 m of oceanwide deposition through time shown for context. The IF record preserves considerably less than this calculation would expect, indicating that, if the IFs were ocean-wide, the entire record has been affected by subduction, deformation, and burial.

References

Alexandrov, E. A. (1973). The Precambrian Banded Iron-Formations of the Soviet Union. *Economic Geology*, 68(7), 1035–1062. https://doi.org/10.2113/gsecongeo.68.7.1035

Allen, R. L., Lundstrom, I., Ripa, M., & Christofferson, H. (1996). Facies analysis of a 1.9 Ga, continental margin, back-arc, felsic caldera province with diverse Zn-Pb-Ag-(Cu-Au) sulfide and

Fe oxide deposits, Bergslagen region, Sweden. *Economic Geology*, 91(6), 979–1008. https://doi.org/10.2113/gsecongeo.91.6.979

- Anbar, A. D., & Knoll, A. H. (2002). Proterozoic Ocean Chemistry and Evolution: A Bioinorganic Bridge? Science, 297(5584), 1137–1142. https://doi.org/10.1126/science.1069651
- Anglin, C., Franklin, J., Loveridge, W., Hunt, P., & Osterberg, S. (1988). Use of zircon U–Pb ages of felsic intrusive and extrusive rocks in eastern Wabigoon Subprovince, Ontario, to place constraints on base metal and gold mineralization (Radiogenic age and isotopic studies: report 2 No. Paper 88-2) (pp. 109–115). Geological Survey of Canada.
- Anhaeusser, C. R. (1976). The geology of the Sheba Hills area of the Barberton Mountain Land, South Africa, with particular reference to the Eureka syncline. *South African Journal of Geology*, 79(2), 253–280.
- Armstrong, R. L. (1981). Radiogenic isotopes: the case for crustal recycling on a near-steady-state nocontinental-growth Earth. *Phil. Trans. R. Soc. Lond. A*, 301(1461), 443–472. https://doi.org/10.1098/rsta.1981.0122
- Arndt, N. T., Nelson, D. R., Compston, W., Trendall, A. F., & Thorne, A. M. (1991). The age of the Fortescue Group, Hamersley Basin, Western Australia, from ion microprobe zircon U -Pb results. *Australian Journal of Earth Sciences*, 38(3), 261–281. https://doi.org/10.1080/08120099108727971
- Babinski, M., Chemale, F., & Van Schmus, W. R. (1995). The Pb/Pb age of the Minas Supergroup carbonate rocks, Quadrilátero Ferrífero, Brazil. *Precambrian Research*, 72(3), 235–245. https://doi.org/10.1016/0301-9268(94)00091-5
- Barley, M. E., Pickard, A. L., & Sylvester, P. J. (1997a). Emplacement of a large igneous province as a possible cause of banded iron formation 2.45 billion years ago. *Nature*, *385*(6611), 55–58. https://doi.org/10.1038/385055a0
- Barley, M. E., Pickard, A. L., & Sylvester, P. J. (1997b). Emplacement of a large igneous province as a possible cause of banded iron formation 2.45 billion years ago. *Nature*, *385*(6611), 55–58. https://doi.org/10.1038/385055a0
- Barton, E. S., Altermann, W., Williams, I. S., & Smith, C. B. (1994). U-Pb zircon age for a tuff in the Campbell Group, Griqualand West Sequence, South Africa: Implications for Early Proterozoic rock accumulation rates. *Geology*, 22(4), 343–346. https://doi.org/10.1130/0091-7613(1994)022<0343:UPZAFA>2.3.CO;2
- Bau, M., Romer, R. L., Lüders, V., & Beukes, N. J. (1999). Pb, O, and C isotopes in silicified Mooidraai dolomite (Transvaal Supergroup, South Africa): implications for the composition of Paleoproterozoic seawater and 'dating' the increase of oxygen in the Precambrian atmosphere. *Earth and Planetary Science Letters*, 174(1), 43–57. https://doi.org/10.1016/S0012-821X(99)00261-7
- Bekker, A., Planavsky, N. J., Krapež, B., Rasmussen, B., Hofmann, A., Slack, J. F., et al. (2014). Iron Formations: Their Origins and Implications for Ancient Seawater Chemistry. *Treatise on Geochemistry (Second Edition)*, 9, 561–628.
- Bekker, A., Planavsky, N., Rasmussen, B., Krapež, B., Hofmann, A., Slack, J., et al. (2014). Iron Formations: Their Origins and Implications for Ancient Seawater Chemistry. In H. Holland & K. Turekian (Eds.), *Treatise on Geochemistry* (Vol. 9, pp. 561–628). Netherlands: Elsevier.
- Bekker, Andrey, Slack, J. F., Planavsky, N., Krapež, B., Hofmann, A., Konhauser, K. O., & Rouxel, O. J. (2010). Iron Formation: The Sedimentary Product of a Complex Interplay among Mantle,

Tectonic, Oceanic, and Biospheric Processes. *Economic Geology*, 105(3), 467–508. https://doi.org/10.2113/gsecongeo.105.3.467

- Beukes, N. J. (1980). Lithofacies and stratigraphy of the Kuruman and Griquatown iron-formations, Northern Cape Province, South Africa. *South African Journal of Geology*, *83*(1), 69–86.
- Beukes, N J, & Gutzmer, J. (2008). Origin and Paleoenvironmental Significance of Major Iron
 Formations at the Archean-Paleoproterozoic Boundary. In S. Hagemann, C. Rosiere, J. Gutzmer,
 & N. Beukes (Eds.), *Banded Iron Formation-Related High-Grade Ore* (Vol. 15, pp. 5–47).
- Beukes, Nicolas J. (1987). Facies relations, depositional environments and diagenesis in a major early Proterozoic stromatolitic carbonate platform to basinal sequence, Campbellrand Subgroup, Transvaal Supergroup, Southern Africa. *Sedimentary Geology*, 54(1–2), 1–46. https://doi.org/10.1016/0037-0738(87)90002-9
- Beukes, Nicolas J., Mukhopadhyay, J., & Gutzmer, J. (2008). Genesis of High-Grade Iron Ores of the Archean Iron Ore Group around Noamundi, India. *Economic Geology*, 103(2), 365–386. https://doi.org/10.2113/gsecongeo.103.2.365
- Bickle, M. J., Martin, A., & Nisbet, E. G. (1975). Basaltic and peridotitic komatiltes and stromatolites above a basal unconformity in the Belingwe greenstone belt, Rhodesia. *Earth and Planetary Science Letters*, 27(2), 155–162. https://doi.org/10.1016/0012-821X(75)90024-2
- Biju-Sekhar, S., Yokoyama, K., Pandit, M. K., Okudaira, T., Yoshida, M., & Santosh, M. (2003). Late Paleoproterozoic magmatism in Delhi Fold Belt, NW India and its implication: evidence from EPMA chemical ages of zircons. *Journal of Asian Earth Sciences*, 22(2), 189–207. https://doi.org/10.1016/S1367-9120(02)00188-8
- Billa, M., Feybesse, J.-L., Bronner, G., Lerouge, C., Milesi, J.-P., Traore, S., & Diaby, S. (1999). Les formations a quartzites rubanes ferrugineux des Monts Nimba et du Simandou □: des unites empilees tectoniquement, sur un soubassement plutonique Archeen (craton de Kenema-Man), lors de l'orogene Eburneen. *Comptes Rendus de l'Academie Des Sciences, Series IIA*, 329, 287–294.
- Bleeker, W., Ketchum, J. W., Jackson, V. A., & Villeneuve, M. E. (1999). The Central Slave Basement Complex, Part I: its structural topology and autochthonous cover. *Canadian Journal of Earth Sciences*, 36(7), 1083–1109. https://doi.org/10.1139/e98-102
- Bowring, S. A., Schmus, W. R. V., & Hoffman, P. F. (1984). U–Pb zircon ages from Athapuscow aulacogen, East Arm of Great Slave Lake, N.W.T., Canada. *Canadian Journal of Earth Sciences*, 21(11), 1315–1324. https://doi.org/10.1139/e84-136
- Bradley, D. C. (2008). Passive margins through earth history. *Earth-Science Reviews*, 91(1), 1–26. https://doi.org/10.1016/j.earscirev.2008.08.001
- Bronner, G., & Chauvel, J. J. (1979). Precambrian banded iron-formations of the Ijil Group (Kediat Ijil, Reguibat Shield, Mauritania). *Economic Geology*, 74(1), 77–94. https://doi.org/10.2113/gsecongeo.74.1.77
- Campbell, I. H., & Hill, R. I. (1988). A two-stage model for the formation of the granite-greenstone terrains of the Kalgoorlie-Norseman area, Western Australia. *Earth and Planetary Science Letters*, 90(1), 11–25. https://doi.org/10.1016/0012-821X(88)90107-0
- Campbell, Ian H., & Davies, D. R. (2017). Raising the continental crust. *Earth and Planetary Science Letters*, 460, 112–122. https://doi.org/10.1016/j.epsl.2016.12.011
- Canfield, D. E. (1998). A new model for Proterozoic ocean chemistry. *Nature*, *396*(6710), 450–453. https://doi.org/10.1038/24839

- Canfield, Donald E., Zhang, S., Wang, H., Wang, X., Zhao, W., Su, J., et al. (2018). A Mesoproterozoic iron formation. *Proceedings of the National Academy of Sciences*, 201720529. https://doi.org/10.1073/pnas.1720529115
- Cannon, W. F., Schulz, K. J., Horton, J. W., & Kring, D. A. (2010). The Sudbury impact layer in the Paleoproterozoic iron ranges of northern Michigan, USA. GSA Bulletin, 122(1–2), 50–75. https://doi.org/10.1130/B26517.1
- Chown, E. H., N'dah, E., & Mueller, W. U. (2000). The relation between iron-formation and low temperature hydrothermal alteration in an Archean volcanic environment. *Precambrian Research*, 101(2), 263–275. https://doi.org/10.1016/S0301-9268(99)00091-1
- Cloud, P. (1973). Paleoecological Significance of the Banded Iron-Formation. *Economic Geology*, 68(7), 1135–1143. https://doi.org/10.2113/gsecongeo.68.7.1135
- Cloud, P. E. (1968). Atmospheric and Hydrospheric Evolution on the Primitive Earth Both secular accretion and biological and geochemical processes have affected earth's volatile envelope. *Science*, *160*(3829), 729–736. https://doi.org/10.1126/science.160.3829.729
- Clout, J. M. F., & Simonson, B. M. (2005). Precambrian iron formations and iron formation-hosted iron ore deposits. *Economic Geology*, 100th Anniversary Volume, 643–679.
- Daly, S. J., Fanning, C. M., & Fairclough, M. C. (1998). Tectonic evolution and exploration potential of the Gawler Craton, South Australia. AGSO Journal of Australian Geology and Geophysics, 17, 145–168.
- Dilek, Y., & Furnes, H. (2014). Ophiolites and Their Origins. *Elements*, 10(2), 93–100. https://doi.org/10.2113/gselements.10.2.93
- Douville, E., Charlou, J. L., Oelkers, E. H., Bienvenu, P., Jove Colon, C. F., Donval, J. P., et al. (2002). The rainbow vent fluids (36°14'N, MAR): the influence of ultramafic rocks and phase separation on trace metal content in Mid-Atlantic Ridge hydrothermal fluids. *Chemical Geology*, *184*(1), 37–48. https://doi.org/10.1016/S0009-2541(01)00351-5
- Eriksson, K. A. (1983). Siliciclastic-hosted iron-formations in the early Archaean Barberton and Pilbara sequences. *Journal of the Geological Society of Australia*, *30*(3–4), 473–482. https://doi.org/10.1080/00167618308729271
- Ewers, W. E., & Morris, R. C. (1981). Studies of the Dales Gorge Member of the Brockman Iron Formation, Western Australia. *Economic Geology*, 76(7), 1929–1953. https://doi.org/10.2113/gsecongeo.76.7.1929
- Fairey, B., Tsikos, H., Corfu, F., & Polteau, S. (2013). U–Pb systematics in carbonates of the Postmasburg Group, Transvaal Supergroup, South Africa: Primary versus metasomatic controls. *Precambrian Research*, 231, 194–205. https://doi.org/10.1016/j.precamres.2013.03.010
- Fedo, C. M., & Eriksson, K. A. (1996). Stratigraphic framework of the ~ 3.0 Ga Buhwa Greenstone Belt: a unique stable-shelf succession in the Zimbabwe Archean Craton. *Precambrian Research*, 77(3), 161–178. https://doi.org/10.1016/0301-9268(95)00053-4
- Findlay, J. M., Parrish, R. R., Birkett, T. C., & Watanabe, D. H. (1995). U–Pb ages from the Nimish Formation and Montagnais glomeroporphyritic gabbro of the central New Québec Orogen, Canada. *Canadian Journal of Earth Sciences*, 32(8), 1208–1220. https://doi.org/10.1139/e95-099
- Fralick, P., & Pufahl, P. K. (2006). Iron Formation in Neoarchean Deltaic Successions and the Microbially Mediated Deposition of Transgressive Systems Tracts. *Journal of Sedimentary Research*, 76(9), 1057–1066. https://doi.org/10.2110/jsr.2006.095

- Fralick, P., Davis, D. W., & Kissin, S. A. (2002). The age of the Gunflint Formation, Ontario, Canada: single zircon U–Pb age determinations from reworked volcanic ash. *Canadian Journal of Earth Sciences*, 39(7), 1085–1091. https://doi.org/10.1139/e02-028
- Frei, R., Dahl, P. S., Duke, E. F., Frei, K. M., Hansen, T. R., Frandsson, M. M., & Jensen, L. A. (2008). Trace element and isotopic characterization of Neoarchean and Paleoproterozoic iron formations in the Black Hills (South Dakota, USA): Assessment of chemical change during 2.9–1.9Ga deposition bracketing the 2.4–2.2Ga first rise of atmospheric oxygen. *Precambrian Research*, 162(3), 441–474. https://doi.org/10.1016/j.precamres.2007.10.005
- Frei, R., Gaucher, C., Poulton, S. W., & Canfield, D. E. (2009). Fluctuations in Precambrian atmospheric oxygenation recorded by chromium isotopes. *Nature*, 461(7261), 250–253. https://doi.org/10.1038/nature08266
- German, C. R., & Seyfried, W. E. (2014). Hydrothermal Processes. In H. D. Holland & K. K. Turekian (Eds.), *Treatise on Geochemistry (Second Edition)* (pp. 191–233). Oxford: Elsevier. https://doi.org/10.1016/B978-0-08-095975-7.00607-0
- Gole, M. J., & Klein, C. (1981). Banded Iron-Formations through Much of Precambrian Time. *The Journal of Geology*, 89(2), 169–183.
- Gross, G. A. (1965). *Geology of Iron Deposits in Canada* (Vol. Vol. 1). Ottawa: Department of Mines and Technical Surveys, Geological Survey of Canada.
- Gross, G. A. (1980). A classification of iron formations based on depositional environments. *The Canadian Mineralogist*, 18(2), 215–222.
- Gumsley, A. P., Chamberlain, K. R., Bleeker, W., Söderlund, U., Kock, M. O. de, Larsson, E. R., & Bekker, A. (2017). Timing and tempo of the Great Oxidation Event. *Proceedings of the National Academy of Sciences*, 114(8), 1811–1816. https://doi.org/10.1073/pnas.1608824114
- Hannah, J. L., Bekker, A., Stein, H. J., Markey, R. J., & Holland, H. D. (2004). Primitive Os and 2316 Ma age for marine shale: implications for Paleoproterozoic glacial events and the rise of atmospheric oxygen. *Earth and Planetary Science Letters*, 225(1), 43–52. https://doi.org/10.1016/j.eps1.2004.06.013
- Hao, J., Sverjensky, D. A., & Hazen, R. M. (2017). A model for late Archean chemical weathering and world average river water. *Earth and Planetary Science Letters*, 457, 191–203. https://doi.org/10.1016/j.epsl.2016.10.021
- Hartmann, L. A., Endo, I., Suita, M. T. F., Santos, J. O. S., Frantz, J. C., Carneiro, M. A., et al. (2006).
 Provenance and age delimitation of Quadrilátero Ferrífero sandstones based on zircon U–Pb isotopes. *Journal of South American Earth Sciences*, 20(4), 273–285. https://doi.org/10.1016/j.jsames.2005.07.015
- Hegner, E., Kröner, A., & Hunt, P. (1994). A precise U Supergroup volcanics in Swaziland. *Journal of African Earth Sciences*, 18(4), 339–341. https://doi.org/10.1016/0899-5362(94)90072-8
- Henry, P. (1995). Étude chimique et isotopique (Nd) de formations ferrifères (banded iron formations ou BIFs) du Craton ouest-africain. *Bulletin Societé Géologique Français*, *166*, 3–13.
- Herzberg, C., Condie, K., & Korenaga, J. (2010). Thermal history of the Earth and its petrological expression. *Earth and Planetary Science Letters*, 292(1), 79–88. https://doi.org/10.1016/j.epsl.2010.01.022

- r Manuscrip Autho
- Hoffman, P F. (1988). United Plates of America, The Birth of a Craton: Early Proterozoic Assembly and Growth of Laurentia. *Annual Review of Earth and Planetary Sciences*, *16*(1), 543–603. https://doi.org/10.1146/annurev.ea.16.050188.002551
- Hoffman, Paul F. (1987). Early Proterozoic Foredeeps, Foredeep Magmatism, and Superior-Type Iron-Formations of the Canadian Shield. In A. Kröner (Ed.), *Proterozic Lithospheric Evolution* (pp. 85–98). American Geophysical Union (AGU). https://doi.org/10.1029/GD017p0085
- Hoffman, Paul F., Bowring, S. A., Buchwaldt, R., & Hildebrand, R. S. (2011). Birthdate for the Coronation paleocean: age of initial rifting in Wopmay orogen, CanadaThis article is one of a series of papers published in this Special Issue on the theme of Geochronology in honour of Tom Krogh. *Canadian Journal of Earth Sciences*, 48(2), 281–293. https://doi.org/10.1139/E10-038
- Holland, H. D. (1973). The Oceans; A Possible Source of Iron in Iron-Formations. *Economic Geology*, 68(7), 1169–1172. https://doi.org/10.2113/gsecongeo.68.7.1169
- Holland, H. D. (2006). The oxygenation of the atmosphere and oceans. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 361(1470), 903–915. https://doi.org/10.1098/rstb.2006.1838
- Horstwood, M. S. A., Nesbitt, R. W., Noble, S. R., & Wilson, J. F. (1999). U-Pb zircon evidence for an extensive early Archean craton in Zimbabwe: A reassessment of the timing of craton formation, stabilization, and growth. *Geology*, 27(8), 707–710. https://doi.org/10.1130/0091-7613(1999)027<0707:UPZEFA>2.3.CO;2
- Hou, K., Li, Y., & Wan, D. (2007). Constraints on the Archean atmospheric oxygen and sulfur cycle from mass-independent sulfur records from Anshan-Benxi BIFs, Liaoning Province, China. Science in China Series D: Earth Sciences, 50(10), 1471–1478. https://doi.org/10.1007/s11430-007-0106-9
- Hurley, P. M., & Rand, J. R. (1969). Pre-Drift Continental Nuclei. *Science*, *164*(3885), 1229–1242. https://doi.org/10.1126/science.164.3885.1229
- Huston, D. L., & Logan, G. A. (2004). Barite, BIFs and bugs: evidence for the evolution of the Earth's early hydrosphere. *Earth and Planetary Science Letters*, 220(1), 41–55. https://doi.org/10.1016/S0012-821X(04)00034-2
- Isachsen, C. E., & Bowring, S. A. (1997). The Bell Lake group and Anton Complex: a basement cover sequence beneath the Archean Yellowknife greenstone belt revealed and implicated in greenstone belt formation. *Canadian Journal of Earth Sciences*, 34(2), 169–189. https://doi.org/10.1139/e17-014
- Isley, A. E. (1995). Hydrothermal Plumes and the Delivery of Iron to Banded Iron Formation. *The Journal of Geology*, *103*(2), 169–185.
- Isley, A. E., & Abbott, D. H. (1999). Plume-related mafic volcanism and the deposition of banded iron formation. *Journal of Geophysical Research: Solid Earth*, 104(B7), 15461–15477. https://doi.org/10.1029/1999JB900066
- Jackson, M. J., Sweet, I. P., & Page, R. W. (1999). South Nicholson basin: Evidence for the early Mesoproterozoic Roper Superbasin. In B. E. Bradshaw & D. L. Scott (Eds.), Integrated Basin Analysis of the Isa Superbasin Using Seismic, Well Log and Geopotential Data: An Evaluation of the Economic Potential of the Northern Lawn Hill Platform. Australian Geological Survey Record.
- James, H. L., & Trendall, A. F. (1982). Banded Iron Formation: Distribution in Time and Paleoenvironmental Significance. In *Mineral Deposits and the Evolution of the Biosphere* (pp. 199–217). Springer, Berlin, Heidelberg. https://doi.org/10.1007/978-3-642-68463-0_11

- James, Harold L. (1983). Chapter 12 Distribution of Banded Iron-Formation in Space and Time. In A. F. Trendall & R. C. Morris (Eds.), *Developments in Precambrian Geology* (Vol. 6, pp. 471–490). Elsevier. https://doi.org/10.1016/S0166-2635(08)70053-7
- James, Harold Lloyd. (1954). Sedimentary facies of iron-formation. *Economic Geology and the Bulletin* of the Society of Economic Geologists, 49(3), 235–293.
- Jordan, T. H. (1988). Structure and Formation of the Continental Tectosphere. *Journal of Petrology*, *Special_Volume*(1), 11–37. https://doi.org/10.1093/petrology/Special_Volume.1.11
- Kalganov, M. I., & Kossovskiy, M. A. (1960). The Kursk magnetic anomalies. *Gosudarstvennoye Izdatel'stvo Geograficheskoy Literatury*, 71.
- Kamo, S. L., & Davis, D. W. (1994). Reassessment of Archean crustal development in the Barberton Mountain Land, South Africa, based on U-Pb dating. *Tectonics*, 13(1), 167–192. https://doi.org/10.1029/93TC02254
- Karlstrom, K. E., Bowring, S. A., & Conway, C. M. (1987). Tectonic significance of an Early Proterozoic two-province boundary in central Arizona. GSA Bulletin, 99(4), 529–538. https://doi.org/10.1130/0016-7606(1987)99<529:TSOAEP>2.0.CO;2
- Kendall, B., Creaser, R. A., Calver, C. R., Raub, T. D., & Evans, D. A. D. (2009). Correlation of Sturtian diamictite successions in southern Australia and northwestern Tasmania by Re–Os black shale geochronology and the ambiguity of "Sturtian"-type diamictite–cap carbonate pairs as chronostratigraphic marker horizons. *Precambrian Research*, 172(3), 301–310. https://doi.org/10.1016/j.precamres.2009.05.001
- Kimberley, M. M. (1978). Paleoenvironmental classification of iron formations. *Economic Geology*, 73(2), 215–229. https://doi.org/10.2113/gsecongeo.73.2.215
- Kinny, P. D., & Nutman, A. P. (1996). Zirconology of the Meeberrie gneiss, Yilgarn Craton, Western Australia: an early Archaean migmatite. *Precambrian Research*, 78(1), 165–178. https://doi.org/10.1016/0301-9268(95)00076-3
- Klein, C. (2005). Some Precambrian Banded Iron-Formations (BIFs) from Around the World: Their Age, Geologic Setting, Mineralogy, Metamorphism, Geochemistry, and Origins. *American Mineralogist*, 90(10), 1473–1499. https://doi.org/10.2138/am.2005.1871
- Klein, C., & Beukes, N. J. (1989). Geochemistry and sedimentology of a facies transition from limestone to iron-formation deposition in the early Proterozoic Transvaal Supergroup, South Africa. *Economic Geology*, 84(7), 1733–1774. https://doi.org/10.2113/gsecongeo.84.7.1733
- Konhauser, K. O., Newman, D. K., & Kappler, A. (2005). The potential significance of microbial Fe(III) reduction during deposition of Precambrian banded iron formations. *Geobiology*, *3*(3), 167–177. https://doi.org/10.1111/j.1472-4669.2005.00055.x
- Konhauser, K. O., Planavsky, N. J., Hardisty, D. S., Robbins, L. J., Warchola, T. J., Haugaard, R., et al. (2017). Iron formations: A global record of Neoarchaean to Palaeoproterozoic environmental history. *Earth-Science Reviews*, 172, 140–177. https://doi.org/10.1016/j.earscirev.2017.06.012
- van Kranendonk, M. J., Hickman, A. H., Smithies, R. H., Nelson, D. R., & Pike, G. (2002). Geology and Tectonic Evolution of the Archean North Pilbara Terrain, Pilbara Craton, Western Australia. *Economic Geology*, 97(4), 695–732. https://doi.org/10.2113/gsecongeo.97.4.695
- Krapez, B., & Martin, D. M. (1999). Sequence stratigraphy of the Palaeoproterozoic Nabberu Province of Western Australia. Australian Journal of Earth Sciences, 46(1), 89–103. https://doi.org/10.1046/j.1440-0952.1999.00692.x

- r Manuscrip Autho
- Krapež, B., Barley, M. E., & Pickard, A. L. (2003). Hydrothermal and resedimented origins of the precursor sediments to banded iron formation: sedimentological evidence from the Early Palaeoproterozoic Brockman Supersequence of Western Australia. *Sedimentology*, 50(5), 979– 1011. https://doi.org/10.1046/j.1365-3091.2003.00594.x
- Kulik, D. A., & Korzhnev, M. N. (1997). Lithological and geochemical evidence of Fe and Mn pathways during deposition of Lower Proterozoic banded iron formation in the Krivoy Rog Basin (Ukraine). *Geological Society, London, Special Publications, 119*(1), 43–80. https://doi.org/10.1144/GSL.SP.1997.119.01.04
- Kump, L. R., & Seyfried, W. E. (2005). Hydrothermal Fe fluxes during the Precambrian: Effect of low oceanic sulfate concentrations and low hydrostatic pressure on the composition of black smokers. *Earth and Planetary Science Letters*, 235(3), 654–662. https://doi.org/10.1016/j.eps1.2005.04.040
- Kusky, T. M. (1998). Tectonic setting and terrane accretion of the Archean Zimbabwe craton. *Geology*, 26(2), 163–166. https://doi.org/10.1130/0091-7613(1998)026<0163:TSATAO>2.3.CO;2
- Laajoki, K. (1975). Rare earth elements in Precambrian iron formations in Väyrylänkylä, south Puolanka Area, Finland. *Bulletin of the Geological Society of Finland*, 47, 91–105.
- Laajoki, K., & Lavikainen, S. (1977). Rare-earth elements in in the Archaean iron formation and associated schists in Ukkolanvaara, Ilomantsi, SE Finland. *Geological Survey of Finland*, *Bulletin*, 49, 105–123.
- Labrosse, S., & Jaupart, C. (2007). Thermal evolution of the Earth: Secular changes and fluctuations of plate characteristics. *Earth and Planetary Science Letters*, 260(3), 465–481. https://doi.org/10.1016/j.epsl.2007.05.046
- Launay, N., Quesnel, Y., Rochette, P., & Demory, F. (2018). Iron Formations as the Source of the West African Magnetic Crustal Anomaly. *Frontiers in Earth Science*, 6. https://doi.org/10.3389/feart.2018.00032
- Luo, G., Ono, S., Beukes, N. J., Wang, D. T., Xie, S., & Summons, R. E. (2016). Rapid oxygenation of Earth's atmosphere 2.33 billion years ago. *Science Advances*, 2(5), e1600134. https://doi.org/10.1126/sciadv.1600134
- Luo, Y., Sun, M., Zhao, G., Li, S., Xu, P., Ye, K., & Xia, X. (2004). LA-ICP-MS U–Pb zircon ages of the Liaohe Group in the Eastern Block of the North China Craton: constraints on the evolution of the Jiao-Liao-Ji Belt. *Precambrian Research*, 134(3), 349–371. https://doi.org/10.1016/j.precamres.2004.07.002
- Machado, N., Lindenmayer, Z., Krogh, T. E., & Lindenmayer, D. (1991). U-Pb geochronology of Archean magmatism and basement reactivation in the Carajás area, Amazon shield, Brazil. *Precambrian Research*, *49*(3), 329–354. https://doi.org/10.1016/0301-9268(91)90040-H
- Machado, N., Schrank, A., Noce, C. M., & Gauthier, G. (1996). Ages of detrital zircon from Archean-Paleoproterozoic sequences: Implications for Greenstone Belt setting and evolution of a Transamazonian foreland basin in Quadrilátero Ferrífero, southeast Brazil. *Earth and Planetary Science Letters*, 141(1), 259–276. https://doi.org/10.1016/0012-821X(96)00054-4
- Maliva, R. G., Knoll, A. H., & Simonson, B. M. (2005). Secular change in the Precambrian silica cycle: Insights from chert petrology. *GSA Bulletin*, 117(7–8), 835–845. https://doi.org/10.1130/B25555.1
- Mapeo, R. B. M., Armstrong, R. A., & Kampunzu, A. B. (2001). SHRIMP U–Pb zircon geochronology of gneisses from the Gweta borehole, northeast Botswana: implications for the Palaeoproterozoic Magondi Belt in southern Africa. *Geological Magazine*, 138(3), 299–308.

- Mapeo, R. B. M., Ramokate, L. V., Armstrong, R. A., & Kampunzu, A. B. (2004). U-Pb zircon age of the upper Palapye group (Botswana) and regional implications. *Journal of African Earth Sciences*, 40(1), 1–16. https://doi.org/10.1016/j.jafrearsci.2004.08.001
- McKenzie, D., & Weiss, N. (1975). Speculations on the Thermal and Tectonic History of the Earth. *Geophysical Journal International*, 42(1), 131–174. https://doi.org/10.1111/j.1365-246X.1975.tb05855.x
- McKenzie, D. P. (1969). Speculations on the Consequences and Causes of Plate Motions. *Geophysical Journal of the Royal Astronomical Society*, *18*(1), 1–32. https://doi.org/10.1111/j.1365-246X.1969.tb00259.x
- Molnar, P., & Gray, D. (1979). Subduction of continental lithosphere: Some constraints and uncertainties. *Geology*, 7(1), 58–62. https://doi.org/10.1130/0091-7613(1979)7<58:SOCLSC>2.0.CO;2
- Moores, E. M. (1982). Origin and emplacement of ophiolites. *Reviews of Geophysics*, 20(4), 735–760. https://doi.org/10.1029/RG020i004p00735
- Morey, G. B., & Southwick, D. L. (1995). Allostratigraphic relationships of early Proterozoic ironformations in the Lake Superior region. *Economic Geology*, 90(7), 1983–1993. https://doi.org/10.2113/gsecongeo.90.7.1983
- Morris, R. C. (1993). Genetic modelling for banded iron-formation of the Hamersley Group, Pilbara Craton, Western Australia. *Precambrian Research*, 60(1), 243–286. https://doi.org/10.1016/0301-9268(93)90051-3
- Mücke, A., Annor, A., & Neumann, U. (1996). The Algoma-type iron-formations of the Nigerian metavolcano-sedimentary schist belts. *Mineralium Deposita*, *31*(1–2), 113–122.
- Mücke, Arno. (2005). The Nigerian manganese-rich iron-formations and their host rocks—from sedimentation to metamorphism. *Journal of African Earth Sciences*, *41*(5), 407–436. https://doi.org/10.1016/j.jafrearsci.2005.07.003
- Mueller, P., Burger, H., Wooden, J., Heatherington, A., Mogk, D., & D'Arcy, K. (2004). Age and evolution of the Precambrian crust of the Tobacco Root Mountains, Montana. In J. Brady, H. Burger, J. Cheney, & T. Harms (Eds.), *Precambrian Geology of the Tobacco Root Mountains, Montana* (pp. 181–202). Boulder, CO: Geological Society of America.
- Mueller, P. A., Wooden, J. L., Nutman, A. P., & Mogk, D. W. (1998). Early Archean crust in the northern Wyoming province: Evidence from U–Pb ages of detrital zircons. *Precambrian Research*, 91(3), 295–307. https://doi.org/10.1016/S0301-9268(98)00055-2
- Mukhopadhyay, J., Beukes, N. J., Armstrong, R. A., Zimmermann, U., Ghosh, G., & Medda, R. A. (2008). Dating the Oldest Greenstone in India: A 3.51-Ga Precise U-Pb SHRIMP Zircon Age for Dacitic Lava of the Southern Iron Ore Group, Singhbhum Craton. *The Journal of Geology*, *116*(5), 449–461. https://doi.org/10.1086/590133
- Müller, P. J., & Suess, E. (1979). Productivity, sedimentation rate, and sedimentary organic matter in the oceans—I. Organic carbon preservation. *Deep Sea Research Part A. Oceanographic Research Papers*, 26(12), 1347–1362. https://doi.org/10.1016/0198-0149(79)90003-7
- Naqvi, S. M., Sawkar, R. H., Subba Rao, D. V., Govil, P. K., & Gnaneswar Rao, T. (1988). Geology, geochemistry and tectonic setting of Archaean greywackes from Karnataka nucleus, India. *Precambrian Research*, 39(3), 193–216. https://doi.org/10.1016/0301-9268(88)90042-3
- Nelson, D. R. (1998). *Compilation of SHRIMP U-Pb zircon geochronology data*, 1997 (No. 1998/2). Western Australian Geological Survey.

- Author Manuscript
- Nielsen, S. G., Rehkämper, M., Teagle, D. A. H., Butterfield, D. A., Alt, J. C., & Halliday, A. N. (2006). Hydrothermal fluid fluxes calculated from the isotopic mass balance of thallium in the ocean crust. *Earth and Planetary Science Letters*, 251(1), 120–133. https://doi.org/10.1016/j.epsl.2006.09.002
- Noce, C. M., Zuccheti, M., Baltazar, O. F., Armstrong, R., Dantas, E., Renger, F. E., & Lobato, L. M. (2005). Age of felsic volcanism and the role of ancient continental crust in the evolution of the Neoarchean Rio das Velhas Greenstone belt (Quadrilátero Ferrífero, Brazil): U–Pb zircon dating of volcaniclastic graywackes. *Precambrian Research*, 141(1), 67–82. https://doi.org/10.1016/j.precamres.2005.08.002
- Nutman, A. P., Chadwick, B., Krishna Rao, B., & Vasudev, V. N. (1996). SHRIMP U/Pb zircon ages of acid volcanic rocks in the Chitradurga and Sandur groups, and granites adjacent to the Sandur Schist belt, Karnataka. *Journal of the Geological Society of India*, 47(2), 153–164.
- Nutman, Allen P., Bennett, V. C., Friend, C. R. L., & Rosing, M. T. (1997). ~ 3710 and ≥ 3790 Ma volcanic sequences in the Isua (Greenland) supracrustal belt; structural and Nd isotope implications. *Chemical Geology*, 141(3), 271–287. https://doi.org/10.1016/S0009-2541(97)00084-3
- Nutman, Allen P., Friend, C. R. L., & Paxton, S. (2009). Detrital zircon sedimentary provenance ages for the Eoarchaean Isua supracrustal belt southern West Greenland: Juxtaposition of an imbricated ca. 3700Ma juvenile arc against an older complex with 3920–3760Ma components. *Precambrian Research*, 172(3), 212–233. https://doi.org/10.1016/j.precamres.2009.03.019
- Oen, I. S. (1987). Rift-related igneous activity and metallogenesis in SW Bergslagen, Sweden. *Precambrian Research*, 35, 367–382. https://doi.org/10.1016/0301-9268(87)90064-7
- Oliveira, E. P., Mello, E. F., & McNaughton, N. (2002). Reconnaissance U–Pb geochronology of Precambrian quartzites from the Caldeirão belt and their basement, NE São Francisco Craton, Bahia, Brazil: implications for the early evolution of the Paleoproterozoic Itabuna–Salvador– Curaçá orogen. *Journal of South American Earth Sciences*, 15(3), 349–362. https://doi.org/10.1016/S0895-9811(02)00039-1
- Peters, S. E., & Husson, J. M. (2018). We Need a Global Comprehensive Stratigraphic Database: Here's a Start. *The Sedimentary Record*, *16*(1), 4–9. https://doi.org/10.2110/sedred.2018.1.4
- Peucat, J. J., Bouhallier, H., Fanning, C. M., & Jayananda, M. (1995). Age of the Holenarsipur Greenstone Belt, Relationships with the Surrounding Gneisses (Karnataka, South India). *The Journal of Geology*, 103(6), 701–710. https://doi.org/10.1086/629789
- Pickard, A. . (2003a). SHRIMP U–Pb zircon ages for the Palaeoproterozoic Kuruman Iron Formation, Northern Cape Province, South Africa: evidence for simultaneous BIF deposition on Kaapvaal and Pilbara Cratons. *Precambrian Research*, 125(3–4), 275–315. https://doi.org/10.1016/S0301-9268(03)00113-X
- Pickard, A. L. (2003b). SHRIMP U–Pb zircon ages for the Palaeoproterozoic Kuruman Iron Formation, Northern Cape Province, South Africa: evidence for simultaneous BIF deposition on Kaapvaal and Pilbara Cratons. *Precambrian Research*, 125(3), 275–315. https://doi.org/10.1016/S0301-9268(03)00113-X
- Poulton, S. W., & Canfield, D. E. (2011). Ferruginous Conditions: A Dominant Feature of the Ocean through Earth's History. *Elements*, 7(2), 107–112. https://doi.org/10.2113/gselements.7.2.107

- Prendergast, M. D. (2004). The Bulawayan Supergroup: a late Archaean passive margin-related large igneous province in the Zimbabwe craton. *Journal of the Geological Society*, *161*(3), 431–445. https://doi.org/10.1144/0016-764902-092
- Rasmussen, B., & Fletcher, I. R. (2002). Indirect dating of mafic intrusions by SHRIMP U–Pb analysis of monazite in contact metamorphosed shale: an example from the Palaeoproterozoic Capricorn Orogen, Western Australia. *Earth and Planetary Science Letters*, 197(3), 287–299. https://doi.org/10.1016/S0012-821X(02)00501-0
- Rasmussen, B., Fletcher, I. R., Muhling, J. R., & Wilde, S. A. (2010). In situ U–Th–Pb geochronology of monazite and xenotime from the Jack Hills belt: Implications for the age of deposition and metamorphism of Hadean zircons. *Precambrian Research*, 180(1), 26–46. https://doi.org/10.1016/j.precamres.2010.03.004
- Roberts, H., Dahl, P., Kelley, S., & Frei, R. (2002). New 207Pb–206Pb and 40Ar–39Ar ages from SW Montana, USA: constraints on the Proterozoic and Archæan tectonic and depositional history of the Wyoming Province. *Precambrian Research*, 117(1), 119–143. https://doi.org/10.1016/S0301-9268(02)00076-1
- Rollinson, H. (2016). Archaean crustal evolution in West Africa: A new synthesis of the Archaean geology in Sierra Leone, Liberia, Guinea and Ivory Coast. *Precambrian Research*, 281, 1–12. https://doi.org/10.1016/j.precamres.2016.05.005
- Rosière, C. A., Heimann, A., Oyhantçabal, P., & Santos, J. O. S. (2018). The Iron Formations of the South American Platform. In *Geology of Southwest Gondwana* (pp. 493–526). Springer, Cham. https://doi.org/10.1007/978-3-319-68920-3_18
- Rosing, M. T., Rose, N. M., Bridgwater, D., & Thomsen, H. S. (1996). Earliest part of Earth's stratigraphic record: A reappraisal of the >3.7 Ga Isua (Greenland) supracrustal sequence. *Geology*, 24(1), 43–46. https://doi.org/10.1130/0091-7613(1996)024<0043:EPOESS>2.3.CO;2
- Rye, R., & Holland, H. D. (1998). Paleosols and the evolution of atmospheric oxygen: a critical review. *American Journal of Science*, 298(8), 621–672.
- Schrank, A., & Silva, M. G. (1993). Greenstone Belts do Cráton do São Francisco, Brasil. In A. Misi & J. M. L. Dominguez (Eds.), O Craton do São Francisco (pp. 85–118). Salvador: SBG/SGM/CNPq.
- Seyfried, W. E., Ding, K., & Berndt, M. E. (1991). Phase equilibria constraints on the chemistry of hot spring fluids at mid-ocean ridges. *Geochimica et Cosmochimica Acta*, 55(12), 3559–3580. https://doi.org/10.1016/0016-7037(91)90056-B
- Shimizu, H., Umemoto, N., Masuda, A., & Appel, P. W. U. (1990). Sources of iron-formations in the archean isua and malene supracrustals, West Greenland: Evidence from La-Ce and sm-nd isotopic data and REE abundances. *Geochimica et Cosmochimica Acta*, 54(4), 1147–1154. https://doi.org/10.1016/0016-7037(90)90445-Q
- Silva, L. C. da, Armstrong, R., Noce, C. M., Carneiro, M. A., Pimentel, M., Pedrosa-Soares, A. C., et al. (2002). Reavaliação da evolução geológica em terrenos pré-cambrianos brasileiros com base em novos dados U–Pb SHRIMP, parte II: orógeno Araçuaí, Cinturão Mineiro e Cráton São Francisco Meridional. *Revista Brasileira de Geociências*, 32(4), 513–528.
- Simonson, B. M. (2003). Origin and evolution of large Precambrian iron formations. In M. A. Chan & A. W. Archer (Eds.), *Extreme Depositional Environments: Mega End Members in Geologic Time* (Special Paper 370, pp. 231–244). Geological Society of America.

- Sleep, N. H., & Bird, D. K. (2008). Evolutionary ecology during the rise of dioxygen in the Earth's atmosphere. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 363(1504), 2651–2664. https://doi.org/10.1098/rstb.2008.0018
- Sleep, N. H., & Windley, B. F. (1982). Archean Plate Tectonics: Constraints and Inferences. *The Journal of Geology*, 90(4), 363–379. https://doi.org/10.1086/628691
- Smith, A. J. B., Beukes, N. J., & Gutzmer, J. (2013). The Composition and Depositional Environments of Mesoarchean Iron Formations of the West Rand Group of the Witwatersrand Supergroup, South Africa. *Economic Geology*, 108(1), 111–134. https://doi.org/10.2113/econgeo.108.1.111
- Stott, G., & Corfu, F. (1991). *Uchi Subprovince* (Geology of Ontario No. Special Volume 4, Part 1) (pp. 145–238). Ontario Geological Survey.
- Sugisaki, R. (1984). Relation between Chemical Composition and Sedimentation Rate of Pacific Ocean-Floor Sediments Deposited since the Middle Cretaceous: Basic Evidence for Chemical Constraints on Depositional Environments of Ancient Sediments. *The Journal of Geology*, 92(3), 235–259.
- Szpunar, M., Hand, M., Barovich, K., Jagodzinski, E., & Belousova, E. (2011). Isotopic and geochemical constraints on the Paleoproterozoic Hutchison Group, southern Australia: Implications for Paleoproterozoic continental reconstructions. *Precambrian Research*, 187(1), 99–126. https://doi.org/10.1016/j.precamres.2011.02.006
- Taner, M. F., & Chemam, M. (2015). Algoma-type banded iron formation (BIF), Abitibi Greenstone belt, Quebec, Canada. Ore Geology Reviews, 70, 31–46. https://doi.org/10.1016/j.oregeorev.2015.03.016
- Thiéblemont, D., Goujou, J. C., Egal, E., Cocherie, A., Delor, C., Lafon, J. M., & Fanning, C. M. (2004). Archean evolution of the Leo Rise and its Eburnean reworking. *Journal of African Earth Sciences*, 39(3), 97–104. https://doi.org/10.1016/j.jafrearsci.2004.07.059
- Trendall, A. F. (2002). The Significance of Iron-Formation in the Precambrian Stratigraphic Record. In W. Altermann & P. L. Corcoran (Eds.), *Precambrian Sedimentary Environments: A Modern Approach to Ancient Depositional Systems* (pp. 33–66). Spec. Publs. int. Ass. Sediment. https://doi.org/10.1002/9781444304312.ch3
- Trendall, A. F., & Blockley, J. G. (1970). The iron formations of the Precambrian Hamersley Group, Western Australia, with special reference to the associated crocidolite (Vol. 119). Geological Survey of Western Australia.
- Trendall, A. F., Nelson, D. R., Thorne, A. M., Compston, W., Williams, I. S., & Armstrong, R. A. (1990). Precise zircon U-Pb chronological comparison of the volcano-sedimentary sequences of the Kaapvaal and Pilbara Cratons between about 3.1 and 2.4 Ga. *Proceedings of the Third International Archaean Symposium, Perth*, 81–83.
- Trendall, A. F., Laeter, J. R. de, Nelson, D. R., & Mukhopadhyay, D. (1997). A precise zircon U-Pb age for the base of the BIF of the Mulaingiri formation, (Bababudan group, Dharwar supergroup) of the Karnataka craton. *Journal of the Geological Society of India*, *50*(2), 161–170.
- Trendall, A. F., Compston, W., Nelson, D. R., Laeter, J. R. D., & Bennett, V. C. (2004). SHRIMP zircon ages constraining the depositional chronology of the Hamersley Group, Western Australia*. *Australian Journal of Earth Sciences*, 51(5), 621–644. https://doi.org/10.1111/j.1400-0952.2004.01082.x

- Author Manuscrip
- Turek, A., Smith, P. E., & Schmus, W. R. V. (1982). Rb–Sr and U–Pb ages of volcanism and granite emplacement in the Michipicoten belt—Wawa, Ontario. *Canadian Journal of Earth Sciences*, 19(8), 1608–1626. https://doi.org/10.1139/e82-138
- Walker, J. C., Klein, C., Schidlowski, J. W., Schopf, J. W., Stevenson, D. J., & Walter, M. R. (1983). Environmental evolution of the Archean-Early Proterozoic Earth. In *Earth's Earliest Biosphere* (pp. 260–290). Princeton, N.J.: Princeton University Press.
- Walker, J. C. G., & Brimblecombe, P. (1985). Iron and Sulfur in the Pre-biologic ocean. *Precambrian Research*, 28, 205–222.
- Wan, Y., Zhang, Q., & Song, T. (2003). SHRIMP ages of detrital zircons from the Changcheng System in the Ming Tombs area, Beijing: Constraints on the protolith nature and maximum depositional age of the Mesoproterozoic cover of the North China Craton. *Chinese Science Bulletin*, 48(22), 2500– 2506. https://doi.org/10.1360/03wd0005
- Wilson, J. P., Fischer, W. W., Johnston, D. T., Knoll, A. H., Grotzinger, J. P., Walter, M. R., et al. (2010). Geobiology of the late Paleoproterozoic Duck Creek Formation, Western Australia. *Precambrian Research*, 179(1), 135–149. https://doi.org/10.1016/j.precamres.2010.02.019
- Zeh, A., Gerdes, A., & Heubeck, C. (2013). U–Pb and Hf isotope data of detrital zircons from the Barberton Greenstone Belt: constraints on provenance and Archaean crustal evolution. *Journal of the Geological Society*, 170(1), 215–223. https://doi.org/10.1144/jgs2011-162





A. Isley, 1995; Isley and Abbott, 1999

This afticle is protected by copy fight. All right free rved. Time (millions of years)







IF mass scaled